



Yip, S. T. H., Biggs, J., & Albino, F. (2019). Reevaluating volcanic deformation using atmospheric corrections: Implications for the magmatic system of Agung volcano, Indonesia. *Geophysical Research Letters*. https://doi.org/10.1029/2019GL085233

Publisher's PDF, also known as Version of record

License (if available): CC BY Link to published version (if available): 10.1029/2019GL085233

Link to publication record in Explore Bristol Research PDF-document

This is the final published version of the article (version of record). It first appeared online via American Geophysical Union (AGU) at https://agupubs.onlinelibrary.wiley.com/doi/full/10.1029/2019GL085233 . Please refer to any applicable terms of use of the publisher.

University of Bristol - Explore Bristol Research General rights

This document is made available in accordance with publisher policies. Please cite only the published version using the reference above. Full terms of use are available: http://www.bristol.ac.uk/pure/about/ebr-terms



ഹി

Geophysical Research Letters

RESEARCH LETTER

10.1029/2019GL085233

Key Points:

- Uplift reported from InSAR at Agung from 2007–2009 is an atmospheric artifact and not related to precursory volcanic unrest
- These observations do not preclude the presence of a magma source deeper than 7.8 km, or recharge at other times
- Reported deformation signals at other tropical volcanoes need to be reexamined

Supporting Information:

- Figure S1
- Figure S2
- Figure S3
- Figure S4
- Figure S5

Correspondence to:

S. T. H. Yip, stanley.th.yip@bristol.ac.uk

Citation:

Yip, S. T. H., Biggs, J., & Albino, F. (2019). Reevaluating volcanic deformation using atmospheric corrections: Implications for the magmatic system of Agung volcano, Indonesia. *Geophysical Research Letters*, 46. https://doi.org/10.1029/ 2019GL085233

Received 31 AUG 2019 Accepted 17 NOV 2019 Accepted article online 27 NOV 2019

©2019. The Author.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Reevaluating Volcanic Deformation Using Atmospheric Corrections: Implications for the Magmatic System of Agung Volcano, Indonesia

Stanley Tze Hou Yip¹, Juliet Biggs¹, and Fabien Albino¹

¹Centre for the Observation and Modelling of Earthquakes, Volcanoes and Tectonics (COMET), School of Earth Sciences, University of Bristol, Bristol, UK

Abstract A major challenge in using satellite interferometry (interferometric synthetic aperture radar) for volcanic monitoring in the tropics is distinguishing volcano deformation from atmospheric noise. We reanalyzed interferometric synthetic aperture radar time series from 2007–2009 from Agung volcano, Indonesia, which had previously been used as evidence for unrest. Using uncorrected data, we find an apparent velocity of 5.0 ± 2.7 cm/yr consistent with previous reports, but we show that this signal is consistent with predictions of atmospheric contributions derived from weather models (European Centre for Medium-Range Weather Forecasts-High-Resolution). Following the correction, we find a velocity of 1.4 ± 4.2 cm/yr and conclude that there was no significant deformation related to the inflation of a shallow magma source from 2007–2009. We discuss the implications for the inferred magma storage system at Agung and consider which other reported signals might have been wrongly attributed to deformation and should be reanalyzed.

Plain Language Summary Space-based volcano monitoring is difficult in the tropics for a number of reasons including cloud cover in optical images and decorrelation between radar images. A major challenge in satellite-based volcano monitoring in the tropics is distinguishing volcano deformation from atmospheric noise. We reanalyzed spaceborne radar data over Agung volcano, Indonesia, where deformation was reported between 2007 and 2009. Using a high-resolution weather model, we show that the apparent velocity signals of about 5 cm/yr observed at Agung were caused by atmospheric noise. We conclude that there are no significant deformation signals related to a shallow magma source from 2007–2009. Finally, we consider the implications for the magma plumbing system beneath Agung.

1. Introduction

Deformation has been reported at over 200 volcanoes globally (Biggs & Pritchard, 2017) and the majority of reported observations are based on satellite interferometric synthetic aperture radar (InSAR). However, atmospheric artifacts can lead to the misinterpretation of volcanic deformation (i.e., Beauducel et al., 2000; Poland & Lu, 2008; Remy et al., 2015), particularly at tropical volcanoes where the water vapor content is high and rapidly varying (Ebmeier et al., 2013). Thus, a fundamental question remains, are all these volcanoes actually deforming? And if they are, what kind of deformation can we resolve with the available satellite data and ancillary products?

The Indonesian volcanic arc is home to 13% of the world's active volcanoes, including Agung, Bali, which produced one of the largest eruptions of the twentieth century in 1963 (Self & Rampino, 2012). Uplift was reported at Agung between July 2007 and February 2009 based on InSAR time series from the Advanced Land Observation Satellite (ALOS) (Chaussard & Amelung, 2012). The rate of uplift was reported as 5.7 cm/yr in Chaussard and Amelung (2012) and 7.8 cm/yr in Chaussard et al. (2013) based on different numbers of InSAR images. The deformation was attributed to magma injection into a closed-vent volcano at a depth of 1.4 km below sea level (Figure 1b).

In September 2017, the Indonesian Center for Volcanology and Geological Hazard Mitigation reported a seismic swarm and fumarole activity (Center for Volcanology and Geological Hazard Mitigation Press Release, https://magma.vsi.esdm.go.id), which was followed by eruption on 21 November 2017 (Syahbana et al., 2019). Between August and November 2017, Sentinel interferograms showed uplift between Agung and its



Figure 1. (a) Topographic relief map of Bali, Indonesia, showing the locations of Agung and Batur volcanoes as red and blue triangles, respectively. Yellow squares show the capital city of each regency. The inset figure shows the location of Agung volcano (red triangle) and Holocene volcanoes (black triangle) in South East Asia (Global Volcanism Program, 2013). Conceptual model of (b) the 2007–2009 uplift at Agung (Chaussard & Amelung, 2012) and (c) magma plumbing system beneath Agung and Batur volcanoes during the 2017 eruption, modified from Albino et al. (2019).

neighboring volcano, Batur. However, atmospheric corrections were required to properly characterize the deformation due to persistent atmospheric artifacts (Albino et al., 2019). The 2017 deformation was interpreted as a dike intrusion at a depth of 7–13 km b.s.l. (Albino et al., 2019) (Figure 1c), different to the 2007–2009 shallow magma source reported by Chaussard and Amelung (2012).

This paper aims to reevaluate InSAR data at Agung for the period 2007–2009 using newly available atmospheric correction techniques and in the light of new InSAR results from Albino et al. (2019). We hypothesize that the apparent deformation signals reported by Lingyun et al. (2013), Chaussard and Amelung (2012) and Chaussard et al. (2013) are caused by topographically correlated atmospheric artifacts, suggesting that Agung was not deforming in 2007–2009. Consequently, the conceptual model of Agung's magmatic system and recharge between the 1963 and 2017 eruptions requires reanalysis.

2. Methods

To reevaluate the reported deformation of Agung, we processed the same interferogram pairs used by Chaussard and Amelung (2012) and Chaussard et al. (2013). The synthetic aperture radar images are acquired by the Japan Aerospace Exploration Agency ALOS Phased-Array L-band Synthetic Aperture Radar instrument. We used the Repeat Orbit Interferometry Package developed by Jet Propulsion Laboratory and the California Institute of Technology (Rosen et al., 2004) to produce 31 interferograms from ALOS ascending Track 7010, Scene 422 spanning 21 February 2007 to 26 February 2009 (Table S1 and Figure S1 in the supporting information). Three pairs used by Chaussard and Amelung (2012) had processing issues and were excluded (Table S1 and Figure S1). While Chaussard and Amelung (2012) used a reference date of 21 February 2007, we chose 9 July 2007 as the reference date for our study because it marks the beginning of the period of uplift from the uncorrected interferograms. The topographic phase was removed from each interferogram using a 30-m Shuttle Radar Topography Mission 1 Digital Elevation Model (Farr et al., 2007). All interferograms were filtered using a power spectrum filter (Goldstein & Werner, 1998) and unwrapped using the branch cut algorithm (Goldstein et al., 1988). The radar phase change was converted to millimeter range





Figure 2. Effect of reference point. The time series is estimated for each possible reference pixel, and the estimated velocity at the summit is plotted at the location of the reference pixel (a) for uncorrected interferograms and (b) for corrected interferograms. *Each LOS velocity pixel corresponds to the coordinate of the reference point (see Figure S2), which is later interpolated to plot this map.* The black boxes around Agung and Batur denote LOS velocity values not included in the frequency distribution. (c) Frequency distribution of LOS velocity at Agung using data before interpolation and excluding values obtained within the black square. Before applying GACOS atmospheric correction, the highest frequency of LOS velocity at Agung is 5.0 ± 0.9 cm/yr. This is similar to the LOS velocity observed by Chaussard and Amelung (2012) at 5.7 cm/yr indicated by the red dash line. After applying GACOS correction (d), the mean LOS velocity at Agung is 1.4 ± 0.8 cm/yr.

change whereby positive measurements correspond to displacement toward the satellite. Long wavelength ramps due to ionosphere or orbital error were removed using linear ramps before atmospheric correction.

We performed a least squares inversion using a Small Baseline Subset approach (i.e., Berardino et al., 2002; Schmidt & Bürgmann, 2003) to generate a time series and a line-of-sight (LOS) velocity map for Agung and Batur. We calculate time series in an 11 × 11 window and plot the mean and standard deviation (2σ) of these values. We applied a linear regression to each time series to measure the LOS velocity with errors bars stated at a 95% confidence interval (2σ) .

InSAR measures relative displacement, meaning the displacement of a point of interest is dependent on the choice of reference point. It is therefore important to select a stable reference point, but this can be especially challenging where there is strong atmospheric noise. One approach is to fit a plane through the nondeforming areas to establish a reference level (i.e., Delgado et al., 2017; Henderson & Pritchard, 2013), but this is challenging for small volcanic islands, where the nondeforming area is small and poorly defined. To quantify the uncertainty associated with the choice of reference pixel at Agung, we repeat the LOS velocity calculation using different reference points on a grid of 93 m \times 90 m, excluding the area around Agung and Batur. We plot the result in a map view whereby each point represents the LOS velocity of Agung measured using that point as the chosen reference pixel (Figure S2). To help visualization, the coarse grid is interpolated using the gap filling method of Garcia (2010) and Wang et al. (2012) and the probability distribution of velocity values plotted (Figures 2a and 2b).

To estimate the atmospheric contribution to our interferograms, we use the Generic Atmospheric Correction Online Service for InSAR (GACOS). GACOS is based on the European Centre for Medium-Range Weather Forecasts High-Resolution weather model (0.125° and 6-hr resolutions) and uses an Iterative Tropospheric Decomposition method to separate stratified tropospheric signals and turbulent signals and interpolates



Figure 3. LOS velocity map of Bali. (a) Uncorrected LOS velocity, (b) prediction of atmospheric model (GACOS), and (c) corrected LOS velocity from 7 July 2007 to 26 February 2009 (see data and methods for details). The dashed lines indicate the profile of LOS velocity signals (see Figure 4), while the black square denotes the reference point. (d) The time series of LOS displacement of InSAR and GACOS of Agung is shown to display the similarity of atmospheric artifacts with InSAR signals. (e) Corrected time series for Agung (red) and Batur (blue). The LOS velocity (filled line) is calculated by applying a linear regression to each time series (Table S3). (f) Comparison between ground velocity and topography that shows a strong correlation between InSAR and GACOS.

them using the 90-m Shuttle Radar Topography Mission Digital Elevation Model (Yu et al., 2017; Yu, Li, & Penna, 2018; Yu, Li, Penna, & Crippa, 2018). We then convert the phase delays of GACOS model from zenith to the LOS of the interferograms.

3. Results

We apply a linear regression to the time series of uncorrected interferograms and find a LOS velocity of 5.0 ± 2.7 cm/yr (Figure 3a), which is slightly lower than that found by Chaussard and Amelung (2012) and Chaussard et al. (2013). By varying the choice of the reference pixel, we find an uncertainty of the central value of ± 0.9 cm/yr (Figure 2b), consistent with the spread of previously reported values. This highlights the importance of the choice of reference pixel when reporting InSAR measurements in noisy atmospheric conditions. For this study, we select a reference pixel coordinate that corresponds to the peak of the histogram (Figure 2c), which is [115.375°E, 8.451°S] (Table S2).

The time series produced using the GACOS model show a similar pattern to the uncorrected time series (Figures 3d and 3e) with rates of 3.6 ± 2.0 and 5.0 ± 2.7 cm/yr, respectively. After applying the GACOS correction, the residual velocity is 1.4 ± 4.2 cm/yr (Figure 3f). The correlation between topography and signal reduces from $R^2 = 0.39$ before correction to $R^2 = 0.02$ after correction, suggesting that the signal in the uncorrected interferograms are largely attributed to topographically correlated atmospheric artifacts rather than real deformation.

Both Agung and Batur have residual velocities on the order of 1.4 cm/yr. We investigate this further by extracting profiles across the summits of Agung and Batur (Figure 4). At both volcanoes, the uncorrected signal shows a clear relationship with topography, with a larger magnitude at the higher-relief Agung. The GACOS model shows a similar pattern, but with a lower magnitude and greater spatial extent, likely as a result of the interpolation of the coarser weather model. Thus, the correction underestimates the peak of the signal and overestimates in the surrounding area producing the characteristic donut shaped spatial pattern seen in the corrected interferograms (Figure 3c).



Figure 4. Velocity profile of (a, b) Agung and (c, d) Batur. LOS velocity signals of uncorrected time series (dashed line), GACOS model (dotted line), corrected time series (filled line), and topography (black line with filled gray area) of Agung and Batur correspond to the dashed lines in Figures 3a–3c, which are 10 km north-south and 10 km west-east across the summit of Agung and Batur. The relief of Batur is not high enough to produce a clear correlation between phase and topography.

4. Discussion

4.1. Importance of Atmospheric Correction in Volcano InSAR

We show that the deformation signals previously reported at Agung correlate with atmospheric artifacts estimated using weather models. This observation is consistent with previous studies, which shows that interferograms at low latitudes (Ebmeier et al., 2013) and close to coastline are prone to the effects of atmospheric noise (i.e., Albino et al., 2019; Bastin et al., 2007; Huang & Wen, 2014). Despite the similarity between the observed signal and the GACOS model, the correction is not perfect, leaving a residual signal of 1.4 cm/yr and increasing the standard deviation from from 2.7 to 4.2 cm/yr. The coarse spatial resolution of European Centre for Medium-Range Weather Forecasts High-Resolution weather model (0.125° resolution) limits the removal of atmospheric artifacts smaller than \approx 13.9 km (Zebker et al., 1997), which may leave residuals in the interferogram. This reflects the importance of using the highest available resolution weather models when correcting InSAR images (Parker et al., 2015).

Here we have demonstrated that reports of deformation at Agung in 2007-2009 were due to a misinterpretation of atmospheric signals, but how many of the 220 other volcanoes with reported deformation are also misclassified? Atmospheric artifacts are dependent on the topography and geographic location of the volcano (Parker et al., 2015) and tropical volcanoes are likely to experience the largest magnitude of artifacts (Ebmeier et al., 2013). Chaussard and Amelung (2012) and Chaussard et al. (2013) used pairwise logic (Massonnet et al., 1995) to argue that the signal at Agung volcano was real uplift. However, while this method is may still be useful in temperate regions, we find it is less useful in tropical regions where water vapor variations are continuous, rather than associated with discrete weather events. Fortunately, many deformation episodes initially detected by satellite have been confirmed by ground-based measurements (i.e., Lloyd et al., 2019; Parks et al., 2015). However, many have not, including the seven other Indonesian volcanoes where deformation was reported by Chaussard and Amelung (2012) and Chaussard et al. (2013). Of these, Lamongan and Anak Krakatau are much lower relief than Agung, and the signals in those volcanoes are unlikely to be attributed to atmospheric artifacts. However, Slamet, Lawu, Merapi, and Sinabung have summit elevations similar or greater than Agung and reanalysis using modern atmospheric correction techniques is warranted. Developing countries in Southeast Asia, such as Indonesia and Philippines, could benefit enormously from the use of satellite InSAR for monitoring volcanoes considering the large number of eruptions that occur and the lack of ground-based networks (Fernández et al., 2017).

Understanding the resemblance of atmospheric anomalies in volcanic deformation is also important for future studies, especially those conducted during volcanic crises where misidentified signals can have more serious consequences. This is particularly challenging as data become more widely available and nonspecialists share results on social media (Loughlin et al., 2015). One possibility is the use of machine learning algorithms, which have the potential to automatically distinguish between atmospheric signals and deformation with minimal expert input (Anantrasirichai et al., 2018, 2019). The output of the machine learning algorithm is the probability that the signal is caused by deformation as opposed to atmospheric artifacts. Thus, the nonexpert user is no longer required to discriminate between atmospheric artifacts and deformation. Unfortunately, it is likely that the restricted acquisition plan of ALOS-2 will continue to limit the study of volcano deformation in this part of the world, until the launch of the open data L-band NISAR mission in 2022.

We also demonstrate the challenges associated with selecting a stable reference point in tropical environments where turbulent atmospheric anomalies are widespread. The differences in the rate of uplift between this study, Chaussard and Amelung (2012) and Chaussard et al. (2013) can be explained by the difference in choice of reference point. Where possible, the best approach is to integrate InSAR data with a ground-based geodetic network to provide an absolute reference frame (i.e., Delouis et al., 2002; Muller et al., 2014). However, in many cases that is not possible, and we propose a method of choosing a reference point by searching through all possible references, looking at the distribution of velocities and choosing a representative point. This is particularly useful for small islands where the nondeforming area is small.

4.2. Magma Plumbing Systems of Volcanoes

We conclude that (1) there is no evidence of significant (>2.7 cm/yr) volcano deformation at Agung from 2007–2009 and (2) there is no evidence of magma recharge into a shallow magma reservoir (<7.8 km). This conclusion requires a reassessment of our conceptual models of the plumbing system and eruptive cycle at Agung.

The lack of evidence for deformation does not disprove the existence of a shallow reservoir. The 1963 (and possibly 2017) eruption of Agung was triggered by magma mingling between a deep basaltic input and an existing basaltic-andesitic to andesitic reservoir (Self & King, 1996), suggesting such reservoirs are at least transiently present (Sparks et al., 2019). Chaussard and Amelung (2012) linked their now discredited shallow deformation source (1.4 km b.s.l.) to a transpressional tectonic regime on the basis that a compressional stress field would prevent shallow magma propagation, (i.e., Cembrano & Lara, 2009; Mériaux & Lister, 2002; Zellmer et al., 2014). In contrast, the 2017 dike intrusion between Agung and Batur indicates an extensional stress field imposed by edifice loading (Albino et al., 2019), suggesting that regional tectonic stresses are less important than the local stress field in guiding magma pathways consistent with observations of large edifice and high flux volcanoes elsewhere (Wadge et al., 2016). The 2017 dike intrusion also demonstrated a previously unrecognized hydraulic connection between the volcanic systems at Agung and Batur, which is consistent with observations of a small eruption at Batur following the 1963 eruption of Agung (Albino et al., 2019; Wheller & Varne, 1986).

Similarly, the lack of measured deformation in 2007–2009 does not discount the possibility that deformation occurred during other time periods. Assuming a linear rate of magma recharge into the same reservoir that erupted in 1963 between 1963 and 2017, to recover the 400 ×10⁶ m³ of dense rock equivalent (Self & Rampino, 2012) would require a constant recharge rate of 7.4×10^6 m³/yr, with the assumption that the eruption was triggered by mass injection or deformation occurs in transient episodic pulses of deformation. For an InSAR detection threshold of 2.7 cm/yr and the assumption of Mogi point source model (Mogi, 1958), this would only be detectable for reservoirs shallower than 7.8 km (Figure S5). Alternatively, if we consider the rate required to accumulate the 23×10^6 m³ associated with the 2017 events (Albino et al., 2019; Syahbana et al., 2019), anything deeper than 1.9 km would not be detected.

However, linear recharge of an elastic reservoir by an incompressible magma is not the only plausible model, and not even the most probable. Given the sparse geodetic record, discrete deformation episodes, exponentially decaying recharge, or incomplete emptying of the magma reservoir by the 1963 eruption or deformation below detection threshold are all plausible. Each of these could lead to an eruption in 2017 without any measurable surface deformation during the 2007–2009 period of L-band InSAR observations. Alternatively, magma input does not necessarily cause significant deformation as viscous deformation and compressible, gas-rich magmas can dissipate the pressure increase associated with magma input or have

viscoelastic response, reducing surface deformation (Gottsmann et al., 2006; Kilbride et al., 2016). A lack of detectable deformation can be caused by a low rate of magma recharge, a high detection threshold, or gas-rich compressible magmas (Ebmeier et al., 2013; Kilbride et al., 2016).

5. Conclusion

We use newly available atmospheric correction techniques to reevaluate the volcanic deformation detected using InSAR at Agung volcano between 2007 and 2009 (Chaussard & Amelung, 2012). After atmospheric correction, we find a LOS velocity of 1.4 ± 4.2 and 1.4 ± 3.1 cm/yr at Agung and Batur, respectively. Our analysis suggests that the previously reported signals were caused by atmospheric artifacts and we conclude that there is no detectable volcano deformation related to a shallow magma source. However, the sparse geodetic data do not preclude other magma recharge scenarios, including steady recharge of a deep magma system (>7.8 km), rapid recharge in the 1960s, or pulses of recharge outside the 2007–2009 observation window. This study shows that monitoring tropical volcanoes using InSAR remains a challenge, and due to the high elevation and its proximity to the ocean, Agung is particularly prone to atmospheric artifacts (i.e., Ebmeier et al., 2013; Parker et al., 2015). However, with the right techniques and using atmospheric corrections where appropriate, spectacular examples of ground deformation at tropical volcanoes can be observed (Albino et al., 2019; Hamling et al., 2019).

References

- Albino, F., Biggs, J., & Syahbana, D. K. (2019). Dyke intrusion between neighbouring arc volcanoes responsible for 2017 pre-eruptive seismic swarm at Agung, Bali. Nature Communications, 10(1), 748. https://doi.org/10.1038/s41467-019-08564-9
- Anantrasirichai, N., Biggs, J., Albino, F., & Bull, D. (2019). A deep learning approach to detecting volcano deformation from satellite imagery using synthetic datasets. *Remote Sensing of Environment*, 230, 111179. https://doi.org/10.1016/j.rse.2019.04.032
- Anantrasirichai, N., Biggs, J., Albino, F., Hill, P., & Bull, D. (2018). Application of machine learning to classification of volcanic deformation in routinely generated InSAR data. *Journal of Geophysical Research: Solid Earth*, 123, 6592–6606. https://doi.org/10.1029/2018JB015911 Bastin, S., Champollion, C., Bock, O., Drobinski, P., & Masson, F. (2007). Diurnal cycle of water vapor as documented by a dense GPS
- network in a coastal area during ESCOMPTE IOP2. Journal of Applied Meteorology and Climatology, 46, 167–182. https://doi.org/10. 1175/JAM2450.1
- Beauducel, F., Briole, P., & Froger, J.-L. (2000). Volcano-wide fringes in ERS synthetic aperture radar interferograms of Etna (1992–1998): Deformation or tropospheric effect? *Journal of Geophysical Research*, *105*(B7), 16,391–16,402. https://doi.org/10.1029/2000JB900095
- Berardino, P., Fornaro, G., Lanari, R., & Sansosti, E. (2002). A new algorithm for surface deformation monitoring based on small baseline differential SAR interferograms. *IEEE Transactions on Geoscience and Remote Sensing*, 40(11), 2375–2383. https://doi.org/10.1109/TGRS. 2002.803792
- Biggs, J., & Pritchard, M. E. (2017). Global volcano monitoring: What does it mean when volcanoes deform? *Elements*, 13(1), 17–22. https://doi.org/10.2113/gselements.13.1.17
- Cembrano, J., & Lara, L. (2009). The link between volcanism and tectonics in the southern volcanic zone of the Chilean Andes: A review. *Tectonophysics*, 471(1-2), 96–113. https://doi.org/10.1016/J.TECTO.2009.02.038

Chaussard, E., & Amelung, F. (2012). Precursory inflation of shallow magma reservoirs at west Sunda volcanoes detected by InSAR. Geophysical Research Letters, 39, L21311. https://doi.org/10.1029/2012GL053817

- Chaussard, E., Amelung, F., & Aoki, Y. (2013). Characterization of open and closed volcanic systems in Indonesia and Mexico using InSAR time series. Journal of Geophysical Research: Solid Earth, 118, 3957–3969. https://doi.org/10.1002/jgrb.50288
- Delgado, F., Pritchard, M. E., Ebmeier, S., González, P., & Lara, L. (2017). Recent unrest (2002–2015) imaged by space geodesy at the highest risk chilean volcanoes: Villarrica, Ilaima, and calbuco (southern andes). Journal of Volcanology and Geothermal Research, 344, 270–288.
- Delouis, B., Giardini, D., Lundgren, P., & Salichon, J. (2002). Joint inversion of InSAR, GPS, teleseismic, and strong-motion data for the spatial and temporal distribution of earthquake slip: Application to the 1999 Izmit Mainshock. Bulletin of the Seismological Society of America, 92(1), 278–299. https://doi.org/10.1785/0120000806
- Ebmeier, S. K., Biggs, J., Mather, T. A., & Amelung, F. (2013). Applicability of InSAR to tropical volcanoes: Insights from Central America. Geological Society, London, Special Publications, 380, 15–37. https://doi.org/10.1144/SP380.2

Farr, T. G., Rosen, P. A., Caro, E., Crippen, R., Duren, R., Hensley, S., et al. (2007). The Shuttle Radar Topography Mission. Reviews of Geophysics, 45, RG2004. https://doi.org/10.1029/2005RG000183

- Fernández, J., Pepe, A., Poland, M. P., & Sigmundsson, F. (2017). Volcano geodesy: Recent developments and future challenges. Journal of Volcanology and Geothermal Research, 344, 1–12. https://doi.org/10.1016/j.jvolgeores.2017.08.006
- Garcia, D. (2010). Robust smoothing of gridded data in one and higher dimensions with missing values. *Computational Statistics & Data Analysis*, 54(4), 1167–1178. https://doi.org/10.1016/J.CSDA.2009.09.020
- Global Volcanism Program (2013). Volcanoes of the World (v. 4.7.6 ed.) Smithsonian Institution. Retrieved from https://volcano.si.edu/list volcano holocene.cfm doi: https://doi.org/10.5479/si.GVP.VOTW4-2013
- Goldstein, R. M., & Werner, C. L. (1998). Radar interferogram filtering for geophysical applications. *Geophysical Research Letters*, 25(21), 4035–4038. https://doi.org/10.1029/1998GL900033
- Goldstein, R. M., Zebker, H. A., & Werner, C. L. (1988). Satellite radar interferometry: Two dimensional phase unwrapping. *Radio Science*, 23(4), 713–720. https://doi.org/10.1029/RS023i004p00713
- Gottsmann, J., Rymer, H., & Berrino, G. (2006). Unrest at the Campi Flegrei caldera (Italy): A critical evaluation of source parameters from geodetic data inversion. *Journal of Volcanology and Geothermal Research*, 150(1-3), 132–145. https://doi.org/10.1016/J.JVOLGEORES. 2005.07.002
- Hamling, I. J., Cevuard, S., & Garaebiti, E. (2019). Large-scale drainage of a complex magmatic system: Observations from the 2018 eruption of Ambrym Volcano, Vanuatu. *Geophysical Research Letters*, 46, 4609–4617. https://doi.org/10.1029/2019GL082606

Acknowledgments

The authors thank JAXA for ALOS PALSAR images archived at the Alaska Satellite Facility (2019) (https://vertex. daac.asf.alaska.edu/). We would also like to thank the University of Newcastle for providing atmospheric correction data through GACOS (2019) (http://ceg-research.ncl.ac.uk/v2/ gacos/). The 30-m SRTM DEM can be downloaded from USGS Digital Data Services (2019) (https://dds.cr.usgs. gov/). The authors declare no conflict of interest. J. B. and F. A. are supported by the U.K. Centre for the Observation and Modelling of Earthquakes Volcanoes and Tectonics and the NERC Innovation Grant NE/S013970/1. J. B. and S. Y. are supported by the Leverhulme Trust. S. Y. was supported by an EPSRC Impact Accelerator Account Internship, We thank the reviewer for providing a detailed and constructive feedback.

Henderson, S., & Pritchard, M. (2013). Decadal volcanic deformation in the central andes volcanic zone revealed by InSAR time series. *Geochemistry, Geophysics, Geosystems, 14*, 1358–1374. https://doi.org/10.1002/ggge.20074

Huang, L., & Wen, X. (2014). Temporal variations of atmospheric water vapor δD 350 and δ180 above an arid artificial oasis cropland in the Heihe River Basin. Journal of Geophysical Research: Atmospheres, 119, 11,456–11,476. https://doi.org/10.1002/2014JD021891

Kilbride, B. M. C., Edmonds, M., & Biggs, J. (2016). Observing eruptions of gas-rich compressible magmas from space. Nature Communications, 7, 1–8. https://doi.org/10.1038/ncomms13744

Lingyun, J., Qingliang, W., & Shanlan, Q. (2013). Present-day deformation of Agung volcano, Indonesia, as determined using SBAS-InSAR. Geodesy and Geodynamics, 4(3), 65–70. https://doi.org/10.3724/SPJ.1246.2013.03065

Lloyd, R., Biggs, J., & Copley, A. (2019). The decade-long Machaze-Zinave after shock sequence in the slowly straining Mozambique Rift. Geophysical Journal International, 217(1), 504–531. https://doi.org/10.1093/gji/ggz033

Loughlin, S. C., Sparks, R. S. J., Sparks, S., Brown, S. K., Jenkins, S. F., & Vye-Brown, C. (2015). Global volcanic hazards and risk. Cambridge: Cambridge University Press.

Massonnet, D., Briole, P., & Arnaud, A. (1995). Deflation of Mount Etna monitored by spaceborne radar interferometry. *Nature*, 375(6532), 567–570. https://doi.org/10.1038/375567a0

Mériaux, C., & Lister, J. R. (2002). Calculation of dike trajectories from volcanic centers. *Journal of Geophysical Research*, 107(B4), 2077. https://doi.org/10.1029/2001JB000436

Mogi, K. (1958). Relations between the eruptions of various volcanoes and the deformations of the ground surfaces around them. Bulletin of the Earthquake Research Institute, 36, 99–134. https://doi.org/10.1016/j.epsl.2004.04.016

Muller, C., del Potro, R., Biggs, J., Gottsmann, J., Ebmeier, S. K., Guillaume, S., et al. (2014). Integrated velocity field from ground and satellite geodetic techniques: application to Arenal volcano. *Geophysical Journal International*, 200(2), 863–879. https://doi.org/10.1093/gji/ggu444

- Parker, A. L., Biggs, J., Walters, R. J., Ebmeier, S. K., Wright, T. J., Teanby, N. A., & Lu, Z. (2015). Systematic assessment of atmospheric uncertainties for InSAR data at volcanic arcs using large-scale atmospheric models: Application to the Cascade volcanoes, United States. *Remote Sensing of Environment*, 170, 102–114. https://doi.org/10.1016/J.RSE.2015.09.003
- Parks, M. M., Moore, J. D. P., Papanikolaou, X., Biggs, J., Mather, T. A., Pyle, D. M., et al. (2015). From quiescence to unrest: 20 years of satellite geodetic measurements at Santorini volcano, Greece. *Journal of Geophysical Research: Solid Earth*, 120, 1309–1328. https://doi. org/10.1002/2014JB011540
- Poland, M. P., & Lu, Z. (2008). Radar interferometry observations of surface displacements during pre- and coeruptive periods at Mount St. Helens, Washington, 1992-2005: Chapter 18 in A volcano rekindled: The renewed eruption of Mount St. Helens, 2004-2006: US Geological Survey. https://doi.org/10.3133/PP175018

Remy, D., Chen, Y., Froger, J. L., Bonvalot, S., Cordoba, L., & Fustos, J. (2015). Revised interpretation of recent InSAR signals observed at Llaima volcano (Chile). *Geophysical Research Letters*, 42, 3870–3879. https://doi.org/10.1002/2015GL063872

Rosen, P. A., Hensley, S., Peltzer, G., & Simons, M. (2004). Updated repeat orbit interferometry package released. *Eos, Transactions American Geophysical Union*, 85(5), 47–47. https://doi.org/10.1029/2004EO050004

Schmidt, D. A., & Bürgmann, R. (2003). Time-dependent land uplift and subsidence in the Santa Clara valley, California, from a large interferometric synthetic aperture radar data set. *Journal of Geophysical Research*, 108(B9), 2416. https://doi.org/10.1029/2002JB002267 Self S. & King A. J. (1006). Petrology and sulfur and oblaring emissions of the 1062 emption of Cumung Aging. Belli Indonesia, Bulletin

Self, S., & King, A. J. (1996). Petrology and sulfur and chlorine emissions of the 1963 eruption of Gunung Agung, Bali, Indonesia. Bulletin of Volcanology, 58(4), 263–285. https://doi.org/10.1007/s004450050139

Self, S., & Rampino, M. R. (2012). The 1963–1964 eruption of Agung volcano (Bali, Indonesia). Bulletin of Volcanology, 74(6), 1521–1536. https://doi.org/10.1007/s00445-012-0615-z

Sparks, R. S. J., Annen, C., Blundy, J. D., Cashman, K. V., Rust, A. C., & Jackson, M. D. (2019). Formation and dynamics of magma reservoirs. https://doi.org/10.1098/rsta.2018.0019

Syahbana, D. K., Kasbani, K., Suantika, G., Prambada, O., Andreas, A. S., Saing, U. B., et al. (2019). The 201719 activity at Mount Agung in Bali (Indonesia): Intense unrest, monitoring, crisis response, evacuation, and eruption. *Scientific Reports*, 9(1), 8848. https://doi.org/ 10.1038/s41598-019-45295-9

Wadge, G., Biggs, J., Lloyd, R., & Kendall, J. M. (2016). Historical volcanism and the state of stress in the east African rift system. Frontiers in Earth Science, 4, 86. https://doi.org/10.3389/feart.2016.00086

- Wang, G., Garcia, D., Liu, Y., de Jeu, R., & Johannes Dolman, A. (2012). A three-dimensional gap filling method for large geophysical datasets: Application to global satellite soil moisture observations. *Environmental Modelling & Software*, 30, 139–142. https://doi.org/10. 1016/J.ENVSOFT.2011.10.015
- Wheller, G. E., & Varne, R. (1986). Genesis of dacitic magmatism at batur volcano, Bali, Indonesia: Implications for the origins of stratovolcano calderas. Journal of Volcanology and Geothermal Research, 28(3-4), 363–378. https://doi.org/10.1016/0377-0273(86)90031-4

Yu, C., Li, Z., & Penna, N. T. (2018). Interferometric synthetic aperture radar atmospheric correction using a GPS-based iterative tropospheric decomposition model. *Remote Sensing of Environment*, 204, 109–121. https://doi.org/10.1016/j.rse.2017.10.038

Yu, C., Li, Z., Penna, N. T., & Crippa, P. (2018). Generic atmospheric correction model for interferometric synthetic aperture radar observations. Journal of Geophysical Research: Solid Earth, 123, 9202–9222. https://doi.org/10.1029/2017JB015305

Yu, C., Penna, N. T., & Li, Z. (2017). Generation of real-time mode high-resolution water vapor fields from GPS observations. Journal of Geophysical Research: Atmospheres, 122, 2008–2025. https://doi.org/10.1002/2016JD025753

- Zebker, H. A., Rosen, P. A., & Hensley, S. (1997). Atmospheric effects in interferometric synthetic aperture radar surface deformation and topographic maps. *Journal of Geophysical Research*, 102(B4), 7547–7563. https://doi.org/10.1029/96JB03804
- Zellmer, G. F., Freymuth, H., Cembrano, J. M., Clavero, J. E., Veloso, E. A. E., & Sielfeld, G. G. (2014). Altered mineral uptake into fresh arc magmas: Insights from UTh isotopes of samples from Andean volcanoes under differential crustal stress regimes. *Geological Society, London, Special Publications*, 385(1), 185–208. https://doi.org/10.1144/SP385.9